Author's response to reviews

Title: Prediction of fetal acidemia in placental abruption

Authors:

Jun Konno (junk0036@yahoo.co.jp)
Yoshio Matsuda (yoshiom2979@gmail.com)
Masaki Ogawa (ogawapen@boshi.twmu.ac.jp)
Minoru Mitani (mitani@obgy.twmu.ac.jp)
Hideo Matsui (matsui@obgy.twmu.ac.jp)

Version: 2 Date: 1 February 2013

Author's response to reviews:

We would like to submit a paper, entitled ‘Prediction of fetal acidemia in placental abruption’, which stresses the importance of nonreassuring fetal status, especially the presence of bradycardia in placental abruption and proposes the predictive score for fetal acidemia in placental abruption. To our knowledge, this study is the first to describe the possibilities for fetal acidemia in placental abruption.

All authors have contributed significantly, and all authors have approved the manuscript. This manuscript is not under simultaneous consideration by any other publication.

We are hopeful that this manuscript will be considered for publication, and look forward to receiving your decision.